Author's response to reviews

Title: Skin Metastases of Esophageal Squamous Cell Carcinoma

Authors:

Gabriela B. Iwanski (g.iwanski@uke.uni-hamburg.de)
Andreas Block (block@uke.uni-hamburg.de)
Gunhild Keller (keller@uke.uni-hamburg.de)
Julia Muench (muench@uke.uni-hamburg.de)
S Claus (claus@bakb.net)
Walter Fiedler (fiedler@uke.uni-hamburg.de)
Carsten Bokemeyer (bokemeyer@uke.uni-hamburg.de)

Version: 2 Date: 26 September 2007

Author's response to reviews:

Submission of the manuscript:

Uncommon Presentation of Common Cancer: Multiple Skin Lesions of Esophageal Squamous Cell Carcinoma

Dear Editor in Chief,

we submit the attached manuscript for consideration for publication as an original article in ‘BMC Case Reports’. We report on a rare case of massive cutaneous metastases of Esophageal Squamous Cell Carcinoma. We believe that our work is of significant interest to the readers of your journal.

All authors have reviewed the manuscript and have approved it for submission. The patient has been informed about the publication and agreed to take a picture of his skin metastases. Our results are original, they have not been published previously, and they are not been submitted to any other journal. There is no conflict of interest for any of the authors or for any other personnel involved in this report.

We look forward to hearing from you.

Yours sincerely,

Gabriela B. Iwanski
(First and corresponding author)